Author's response to reviews

Title: Disappointment and adherence among parents to newborns allocated to the control group: A qualitative study of a randomized clinical trial

Authors:

Sandra M Petersen (sandrameinich@gmail.com)
Vibeke Zoffmann (VZO@steno.dk)
Jesper Kjærgaard (jesper.kjaergaard@regionh.dk)
Lone G Stensballe (lone.graff.stensballe@regionh.dk)
Gorm Greisen (gorm.greisen@regionh.dk)

Version: 4 Date: 14 March 2014

Author's response to reviews: see over
Cover letter

Dear Claire Snowdon, Associate Editor of Trials

Thank you for considering the newest revision of our article - *Disappointment and adherence among parents to newborns allocated to the control group: A qualitative study of a randomized clinical trial.*

We are very delighted to hear that you think that we are almost ready to proceed to publication. We have addressed your query and the two minor details (sorry, we should have seen those ourselves!). The changes are marked with track changes in the manuscript.

Once again, I would like to point out how very amazed we are with the effort you have out into revising our manuscript, and we would like to ask, if it would be ok with you that we mention you in the acknowledgements? We will write: “Thanks to Claire Snowdon, BA MA Ph.D., for her very useful and inspiring comments to our manuscript.” Please comment on the acknowledgement if you think that it should be differently stated.

We are looking very much forward to hearing from you.

With best wishes,

Sandra Meinich Petersen, sandrameinich@gmail.com